\begin{table}[ht!]
	\centering
    \caption{The benchmark's settings which were used to analyze the latency of messages sent whilst both the sending and receiving node were busy polling.}
	\label{tab:oneway_settings_busy}
	\vspace{3mm}

    \footnotesize
     \begin{tabular}{!{\hspace{-2pt}}c|!{\hspace{-8pt}}c!{\hspace{-4pt}}c!{\hspace{-4pt}}c!{\hspace{-4pt}}c!{\hspace{-6pt}}c!{\hspace{-2pt}}ccc!{\hspace{-3pt}}ccc!{\hspace{-5pt}}} 
         \multicolumn{1}{c}{} & \rot{service type} & \rot{polling (send)} & \rot{polling (recv)} & \rot{inline mode} & \rot{unsignaled}& \rot{operation} & \rot{burst size} & \rot{repetitions} & \rot{loop pauses} & \rot{timestamp} & \rotright{message size}\\
	 \hlineB{3}
	 \Tstrut
         \autoref{fig:oneway_busy_a} & \graycellleftline{\acrshort{rc}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{20} & \graycell{\SI{0}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\                                                                                        
                                                                                                      
	 \Tstrut                                                                                          
         \autoref{fig:oneway_busy_b} & \graycellleftline{\acrshort{uc}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{20} & \graycell{\SI{0}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\                                                                                        
                                                                                                      
	 \Tstrut                                                                                          
         \autoref{fig:oneway_busy_c} & \graycellleftline{\acrshort{ud}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{20} & \graycell{\SI{0}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\                                                                                        
                                                                                                      
	 \Tstrut                                                                                          
         \autoref{fig:oneway_busy_d} & \graycellleftline{\acrshort{rc}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{1} & \graycell{\SI[scientific-notation=true]{1e9}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\                                                                                        
                                                                                                      
	 \Tstrut                                                                                          
         \autoref{fig:oneway_busy_e} & \graycellleftline{\acrshort{uc}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{1} & \graycell{\SI[scientific-notation=true]{1e9}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\                                                                                        
                                                                                                      
	 \Tstrut                                                                                          
         \autoref{fig:oneway_busy_f} & \graycellleftline{\acrshort{ud}} & busy & busy & \NO & \enspace\NO & send & 8000 & \graycell{1} & \graycell{\SI[scientific-notation=true]{1e9}{\nano\second}} & $t_{subm}$ & \SI{32}{\byte}
	 \Bstrut\\
	\hlineB{3}
	\end{tabular}
\newpage
\end{table}
